Faculty of Science and Engineering

School of Geography, Earth and Environmental Sciences

2018-03-25

The Role of External Inputs and Internal Cycling in Shaping the Global Ocean Cobalt Distribution: Insights From the First Cobalt Biogeochemical Model

Tagliabue, A

http://hdl.handle.net/10026.1/11349

10.1002/2017GB005830 Global Biogeochemical Cycles American Geophysical Union (AGU)

All content in PEARL is protected by copyright law. Author manuscripts are made available in accordance with publisher policies. Please cite only the published version using the details provided on the item record or document. In the absence of an open licence (e.g. Creative Commons), permissions for further reuse of content should be sought from the publisher or author.



Global Biogeochemical Cycles

RESEARCH ARTICLE

10.1002/2017GB005830

Key Points:

- The first global model of ocean cobalt cycling is presented
- Sediments are the major global source of cobalt
- Suppression of Co scavenging by low oxygen and reduced bacterial activity is key in extending the residence time of externally supplied Co

Correspondence to:

A. Tagliabue, a.tagliabue@liverpool.ac.uk

Citation:

Tagliabue, A., Hawco, N. J., Bundy, R. M., Landing, W. M., Milne, A., Morton, P. L., & Saito, M. A. (2018). The role of external inputs and internal cycling in shaping the global ocean cobalt distribution: Insights from the first cobalt biogeochemical model. *Global Biogeochemical Cycles*, 32. https://doi.org/10.1002/ 2017GB005830

Received 31 OCT 2017 Accepted 12 MAR 2018 Accepted article online 25 MAR 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

The Role of External Inputs and Internal Cycling in Shaping the Global Ocean Cobalt Distribution: Insights From the First Cobalt Biogeochemical Model

Alessandro Tagliabue¹, Nicholas J. Hawco^{2,3}, Randelle M. Bundy^{4,5}, William M. Landing⁶, Angela Milne^{6,7}, Peter L. Morton^{6,8}, and Mak A. Saito⁴

¹School of Environmental Sciences, University of Liverpool, Liverpool, UK, ²MIT-WHOI Joint Program in Oceanography/ Applied Ocean Science and Engineering, Department of Marine Chemistry and Geochemistry, Woods Hole Oceanographic Institution, Woods Hole, MA, USA, ³Department of Earth Sciences, University of Southern California, Los Angeles, CA, USA, ⁴Marine Chemistry and Geochemistry, Woods Hole Oceanographic Institution, Woods Hole, MA, USA, ⁵School of Oceanography, University of Washington, Seattle, WA, USA, ⁶Department of Earth, Ocean, and Atmospheric Science, Florida State University, Tallahassee, FL, USA, ⁷School of Geography, Earth and Environmental Sciences, University of Plymouth, Plymouth, UK, ⁸Geochemistry, National High Magnetic Field Laboratory, Tallahassee, FL, USA

Abstract Cobalt is an important micronutrient for ocean microbes as it is present in vitamin B₁₂ and is a co-factor in various metalloenzymes that catalyze cellular processes. Moreover, when seawater availability of cobalt is compared to biological demands, cobalt emerges as being depleted in seawater, pointing to a potentially important limiting role. To properly account for the potential biological role for cobalt, there is therefore a need to understand the processes driving the biogeochemical cycling of cobalt and, in particular, the balance between external inputs and internal cycling. To do so, we developed the first cobalt model within a state-of-the-art three-dimensional global ocean biogeochemical model. Overall, our model does a good job in reproducing measurements with a correlation coefficient of >0.7 in the surface and >0.5 at depth. We find that continental margins are the dominant source of cobalt, with a crucial role played by supply under low bottom-water oxygen conditions. The basin-scale distribution of cobalt supplied from margins is facilitated by the activity of manganese-oxidizing bacteria being suppressed under low oxygen and low temperatures, which extends the residence time of cobalt. Overall, we find a residence time of 7 and 250 years in the upper 250 m and global ocean, respectively. Importantly, we find that the dominant internal resupply process switches from regeneration and recycling of particulate cobalt to dissolution of scavenged cobalt between the upper ocean and the ocean interior. Our model highlights key regions of the ocean where biological activity may be most sensitive to cobalt availability.

Plain Language Summary Biological activity in the sea requires cobalt, primarily due to its role in vitamin B12 but also because it is required in other cellular enzymes. While our observations of cobalt distributions in the ocean is growing, we do not have a quantitative understanding of the role of different external sources of cobalt or how it is internally processed by different biological and chemical processes in the ocean. To answer these questions, we built the first ever global ocean cobalt model that coupled the cycling of cobalt to the major biogeochemical processes occurring in the ocean. Using this model, we identified that sediments are the major cobalt source and that the combination of oxygen levels and scavenging removal by bacteria allow externally supplied cobalt to pervade the ocean as a whole. We find that in certain regions of the upper ocean, cobalt levels may be low enough to affect biological activity but that to quantify this requires further work on how we represent cellular biochemistry.

1. Introduction

When compared to typical phytoplankton requirements, cobalt (Co) emerges as being relatively depleted in seawater (Moore et al., 2013; Saito et al., 2008), and in some ocean regions, there is evidence that Co is the primary or secondary limiting nutrient (Bertrand et al., 2007, 2015; Browning et al., 2017; Sañudo-Wilhelmy et al., 2006). Marine phytoplankton requires Co due to its presence in vitamin B_{12} (also known as cobalamin) and due to its role as a potential cofactor of carbonic anhydrase and alkaline phosphatase, which catalyze carbon fixation and organic phosphorus acquisition, respectively (although a confirmation awaits of a marine pelagic microbe with Co within its alkaline phosphatase; Wojciechowski et al., 2002). In eukaryotic algae, zinc

(Zn) can be substituted for Co in carbonic anhydrase and alkaline phosphatase when Co levels are low (Morel et al., 1994; Sunda & Huntsman, 1995), but cyanobacteria are known to have an obligate Co requirement (Saito et al., 2002). Despite the Co-containing vitamin B_{12} playing a fundamental role in the development of the first theories for algal nutrient limitation and its representation in numerical models (Droop, 1973, 1974), Co cycling remains ignored in contemporary global ocean biogeochemical models.

While early pioneering studies considered Co to be a "scavenged-type" element (Jickells & Burton, 1988), subsequent investigations with lower detection limits identified the nutrient-like depletion in the upper photic zone highlighting the role of biological Co uptake (Martin et al., 1989; Noble et al., 2008; Saito & Moffett, 2002). Thus, Co is better described as a "hybrid-type" element, with external inputs from continental margins (in particular within the major oxygen minimum zones), as well as riverine, and dust sources (Noble et al., 2012; Saito & Moffett, 2002; Saito et al., 2004; Shelley et al., 2012; Zhang et al., 1990). In addition to these external inputs, Co is strongly removed from the dissolved fraction via scavenging (Bruland et al., 2014; Moffett & Ho, 1996). The scavenging of Co is catalyzed by manganese (Mn) oxidizing bacteria, due to the similar ionic radii and redox potentials of Mn and Co (Cowen & Bruland, 1985; Moffett & Ho, 1996; Sunda & Huntsman, 1988).

Like many bioactive metals, Co is often found strongly bound by natural ligands, particularly in the upper water column (Ellwood & van den Berg, 2001; Saito & Moffett, 2001). These ligands have been shown to be produced from cyanobacteria blooms and released to the ocean upon cell lysis (Saito et al., 2005) and are hypothesized to be important in protecting Co from scavenging and in decreasing Co bioavailability to some phytoplankton when complexed (Moffett & Ho, 1996; Saito et al., 2002). Yet unlike other bioactive metals (e.g., iron), in some regions, Co can also be found to be unsaturated by natural ligands, particularly in coastal and polar regions. This results in measurable concentrations of labile Co that are likely far more bioavailable (Saito et al., 2004, 2010). Thus far, the only known Co ligands are likely to be the cobalamins that are known to be synthesized by some bacteria (including cyanobacteria) and archaea.

With the arrival of GEOTRACES research cruises at the ocean basin scale, a number of studies have provided detailed data on the distributions of Co in the Atlantic, Pacific, and Southern Oceans. The zonal sections identified major plumes of Co in the major oxygen-depleted regions of the North and South Atlantic and South Pacific Oceans, extending well into each basin and farther than analogous iron or Mn plumes (Hawco et al., 2016; Noble et al., 2012, 2017), implying lesser Co scavenging. In addition, many of these studies have observed widespread depletions of dissolved (dCo) in the upper ocean, typical of its role as a micronutrient, even in the dust-laden North Atlantic Ocean (Bown et al., 2011; Dulaquais et al., 2014; Hawco et al., 2016; Noble et al., 2012, 2017). These large-scale oceanographic Co data sets have facilitated the examination of the links between Co and other parameters, such as positive links with phosphate and nitrous oxide within the euphotic zone and inverse relationships with dissolved oxygen in the mesopelagic (Noble et al., 2012, 2017). When compared to phosphate, Co is distinguished from other bioactive metals (e.g., cadmium or zinc) in having a wide range of Co:P slopes that span more than an order of magnitude (Bown et al., 2011; Noble et al., 2017; Saito et al., 2010; Saito & Moffett, 2002), suggestive of variable microbial use. In certain ocean regions, Co section data sets spanned regions known to contain hydrothermal plumes of iron and Mn (Hatta et al., 2015; Resing et al., 2015; Saito et al., 2013), but no corresponding Co plumes were observed (Hawco et al., 2016; Noble et al., 2012, 2017).

At the global scale, ocean biogeochemical models are excellent platforms with which to explore the differing roles of often competing signals linked to external inputs and internal cycling in different biogeochemical regimes and water masses. In this study, we developed the first representation of Co cycling in a global ocean model and investigated how external inputs and internal cycling shape the oceanic distribution of this important micronutrient. We find key roles played by low oxygen and the suppression of bacterial activity by temperature in promoting the longevity of Co. This then facilitates the widespread impact of Co inputs from the ocean margins at the basin scale.

2. Methods

We have devised the first global Co model that is coupled to the PISCES-v2 model, which is itself coupled to offline circulation fields within the NEMO framework (www.nemo-ocean.eu). The PISCES-v2 model simulates a wide range of tracers: nitrate, ammonium, phosphate, silicic acid, iron, iron-binding ligands, dissolved oxygen, two size classes of particles, two phytoplankton functional types (diatoms and nanophytoplankton), two



Parameter	Value	Units	Description		
O _{2thres_1}	50	μM O ₂	Threshold for enhanced sedimentary Co fluxes		
O _{2thres_2}	2	μΜ Ο ₂	Threshold for eliminated sedimentary Co fluxes		
θ_{MAX}	150/150	μmol Co/mol P	Maximum phytoplankton Co quotas		
kdCo	50/80	pM Co	Half saturation constants for Co uptake		
KZnCo	0.5	nM Zn	Half saturation constant for Zn-Co interaction for diatom group only		
ΛCo _{MIN}	0.1×10^{-3}	day^{-1}	Minimum dCo scavenging rate		
ΛСο	0.01	day ⁻¹	Maximum dCo scavenging rate		
0 _{2ST}	100	μM O ₂	Threshold for O ₂ effect on scavenging		
kO ₂ Λ	25	$\mu M O_2$	Half saturation constant for O ₂ effect on scavenging and dissolution		
kBΛ	2.5	μMC	Half saturation constant for bacterial effect on scavenging		
kPARΛ	15	W/m ²	Half saturation constant for PAR effect on scavenging		
O _{2DT}	50	μΜ Ο ₂	Threshold for O ₂ effect on sCo dissolution		
λμαχ	0.1	day ⁻¹	Maximum sCo dissolution rate		
CoL _{MIN}	25	pM	Minimum concentration of Co ligands		
θ_{ZOO}	20	μmol Co/mol P	Zooplankton Co quota		
ϕ	1.5	Unitless	Relative rate of particulate organic Co remineralization		

 Table 1

 Model Parameter Values

Note. Where two values are given, the first is for nanophytoplankton and the second is for diatoms.

grazers, dissolved organic carbon, dissolved inorganic carbon, biogenic silica, calcium carbonate, and alkalinity (Aumont et al., 2015; Tagliabue & Resing, 2016). In this work, we have augmented PISCES-v2 with an additional six tracers to resolve the biogeochemical cycling of Co. The additional tracers are dissolved cobalt (dCo), scavenged cobalt (scCo; putatively associated with Mn oxides), cobalt within diatoms (PhyCo_D), cobalt within nanophytoplankton (PhyCo_N), small particulate organic cobalt (PCo_S), and large particulate organic cobalt (PCo_B). The Co within microzooplankton and mesozooplankton is an inferred quantity driven by a fixed Co/P ratio within zooplankton, which then drives excretion of Co when prey Co is greater than the required Co, as for Fe in PISCES-v2 (Aumont et al., 2015). All parameter values for the Co model are described in Table 1.

2.1. Generalized Source-Sink Equations for Co Tracers

$$\frac{d(dCo)}{dt} = \text{Dust}_{Co} + \text{Sed}_{Co} + \text{River}_{Co} - \text{Up}_{Co} - \text{Scav}_{Co} + \text{Dissol}_{Co} + \text{Excret}_{Co} + \text{Remin}_{Co}$$
(1)

dCo is supplied from dust (Dust_{Co}), sediments (Sed_{Co}), and rivers (River_{Co}), with no source from hydrothermal venting based on results from GEOTRACES sections. dCo is consumed by phytoplankton (Up_{Co}) and lost due to scavenging (Scav_{Co}). dCo is also resupplied from the dissolution of scCo (Dissol_{Co}) and excretion by zoo-plankton (Excret_{Co}) and the remineralization of particulate organic cobalt (Remin_{Co}).

$$\frac{d(scCo)}{dt} = Scav_{Co} - Dissol_{Co} - sinking$$
 (2)

scCo is produced due to the scavenging of dCo (Scav_{Co}), and scCo dissolves via dissolution back to dCo (Dissol_{Co}). scCo sinks at 1 m/day.

$$\left. d(phyCo_i) \right|_{dt} = Up_{Coi} - SMS_i$$
 (3)

Phytoplankton Co of functional group *i* (D = diatom or N = nanophytoplankton) is a result of dCo uptake (Up_{Co}) , which is explicitly modeled and specific to diatoms and nanophytoplanktons (Up_{CoD}) and Up_{CoN} , respectively; Up_{Co} in equation (1) is the sum of both terms). Loss of phytoplankton Co follows the processes of mortality, aggregation, and grazing (SMS_i) in the main PISCES-v2 model (Aumont et al., 2015).

$$\frac{d(PCo_i)}{dt} = SMS_i - Remin_{Co}$$
(4)

Remineralization of Co from small and large organic Co particles (subscript i = S or B, Remin_{CoS} and Remin_{CoB}, respectively; Remin_{Co} is the sum of both terms) is modeled independently via a tunable parameter relative to

the remineralization of organic carbon (ϕ). By default, we assume particulate organic Co remineralizes 50% faster than organic carbon (ϕ = 1.5). Other gains and loss of particulate Co (SMS_S and SMS_B) follow the processes of aggregation, disaggregation, phytoplankton and zooplankton mortality, sinking, and grazing as per the main PISCES model (Aumont et al., 2015).

2.2. External Inputs of Co

Dust input assumes a mineral fraction of Co of 17.3 μ g/g (Rudnick & Gao, 2014) and a Co solubility of 8% (Shelley et al., 2012). River supply assumes a Co/C ratio of 12 μ mol/mol (Gaillardet et al., 2003). Sediment Co input (sed_{Co}) is modeled via a set of bottom water oxygen (O_{2bw}) dependent thresholds and is benchmarked to the sedimentary iron (Fe) supply (sed_{Fe}) in the model, which is a function of organic carbon flux to the sediment:

$$\begin{split} &O_{2bw} \leq &O_{2thres_1}, sed_{Co} = sed_{Fe} \times M \times 1,000 \\ &O_{2bw} \leq &O_{2thres_2}, sed_{Co} = 0 \\ &O_{2bw} > &O_{2thres_1}, sed_{Co} = sed_{Fe} \times M \times 25 \end{split}$$

where *M* represents the Co/Fe mineral fraction ratio (Rudnick & Gao, 2014). The 1,000 and 25 factors are tuned to account for suboxic Co release from dissolution of Fe and Mn oxides under suboxic conditions and Co incorporation into Fe sulfides (pyrite) under very low O₂ when sulfate reduction initiates in porewaters of surface sediments. Specifically, $O_{2thres_1} = 50 \ \mu M O_2$ and $O_{2thres_2} = 2 \ \mu M O_2$.

In oxic sediments, Co in porewaters is very low with <2.5 nM (Heggie and Lewis, 1984), due to rapid Mn oxidation in near-surface sediments where O_2 from the water column can penetrate. Oxidation of Co in microbially catalyzed Mn oxidation (Lee & Fisher, 1993; Moffett & Ho, 1996) limits the diffusive flux out of sediments substantially. Co/Al ratios in continental margin sediments (e.g., South China Sea; Hu et al., 2012; 2013) reflects crustal Co/Al ratios, likely because most of the Co released by dissolution/weathering/desorption of crustal materials is returned to the sediments via Mn oxidation in estuaries and coastal seas (i.e., is scavenged; Hawco et al., 2016; Moffett & Ho, 1996).

The small flux when $O_2 > O_{2_thres_1}$ reflects release by sediment desorption and/or ligand stabilization of Co in estuaries (Bewers & Yeats, 1977; Kharkar et al., 1968; Zhang et al., 1990). Suboxic release of Co has been shown explicitly (Johnson et al., 1988; Sundby et al., 1986) and is reflected in low Co/Al ratios in margin sediments in OMZs off Peru (Böning et al., 2004), Chile (Böning et al., 2009), the Gulf of California (Brumsack, 1989), and in the South Atlantic under the Benguela upwelling region (Bremner & Willis, 1993). Co fixation into pyrite formation follows thermodynamic predictions (e.g., Morse & Luther, 1999; Saito et al., 2003) and can be seen from high Co/Al ratios in Black Sea sediments (Brumsack, 2006) and in sulfide-rich sediments near Walvis Bay on the Namibian Coast (Borchers et al., 2005). The precise choice of the threshold concentrations chosen here reflects model tuning to the dissolved oxygen concentrations in the model, balancing model performance in the Atlantic, Pacific, and Indian Oceans.

2.3. Internal Cycling of Co

2.3.1. Phytoplankton Uptake

Phytoplankton uptake of Co is explicitly modeled rather than using a "Redfield" conversion based on the modeled carbon fixation rate. Equation (5) represents this in a similar manner to how PISCES models Fe uptake, accounting for a maximum cellular quota and the potential for uptake to be upregulated under certain conditions (Aumont et al., 2015). This decouples Co uptake from C fixation and permits variable Co/C ratios as observed (Sunda & Huntsman, 1995).

$$Up_{Coi} = \mu_{MAXi} \theta_{MAXi} bCo/(bCo + kbCoi) \frac{(1 - \theta i/\theta_{MAXi})}{(1.05 - \theta i/\theta_{MAXi})} \zeta Zn$$
(5)

where subscript *i* denotes either D (diatoms) or N (nanophytoplankton), bCo is the bioavailable Co pool and is assumed to represent dCo for nanophytoplankton (based on observations/assumptions that cyanobacteria utilize both CoL complexes and Co'; Saito et al., 2002) and inorganic cobalt species, Co', for diatoms (based on observations by Sunda & Huntsman, 1995; see below for the calculation of Co'), and μ_{MAXi} is the maximum growth rate of functional type *i*. The Co/P ratio is represented by θ_i within the



functional group *i*, and θ_{MAXi} is the maximum Co/P ratio for phytoplankton functional group *i*, while kbCo*i* is the half saturation constant for bCo uptake for functional group *i*. Co uptake is downregulated when θ approaches θ_{MAX} using a hyperbolic function with a shape factor set to 0.5 (as for Fe in PISCES-v2). The term ξ_{Zn} is a scalar active only for diatoms and represents the interreplacement of Co and Zn within carbonic anhydrase causing Zn concentrations to affect Co uptake (Price & Morel, 1990; Saito & Goepfert, 2008; Sunda & Huntsman, 1995; Xu et al., 2007) via

$$\xi_{Zn} = MAX \left[0.1, \ 3 \times \left(1 - \frac{Zn}{(Zn + kZnCo)} \right) \right]$$
(6)

where kZnCo represents the half saturation constant for Zn-Co interactions. This is initially set to 0.5 nM Zn, which would approximately reflect a free Zn concentration of 5 pM. Previous work has shown enhanced Co uptake when Zn falls below 5–10 pM in several species of eukaryotic phytoplankton (Sunda & Huntsman, 1995). In the absence of a specific Zn model, the Zn concentration (in nM) is derived from Si (in μ M) using 0.065 × Si + 0.183 (M. C. Lohan, personal communication, 2017) as there is a long noted relationship between Zn and Si (Bruland et al., 1978). The constants 0.1 and 1.3 in equation (6) decrease Co uptake by up to 90% when Zn is abundant and increase Co uptake up to threefold when Zn is scarce, respectively.

2.3.2. Scavenging and Dissolution

The scavenging of dCo is assumed to be driven by Mn oxides produced by Mn-oxidizing bacteria (Johnson et al., 1988; Moffett & Ho, 1996). At this stage, our model does not include an explicit Mn module, so we assume Mn oxides to be prevalent where oxygen is abundant (Ohnemus et al., 2017; Ohnemus & Lam, 2015) and that the activity of heterotrophic Mn-oxidizing bacteria scales with total bacterial activity (Cowen & Bruland, 1985; Moffett & Ho, 1996; Sunda & Huntsman, 1988), except in the surface ocean where manganese oxides are destroyed via photoreduction and dissolution (Sunda & Huntsman, 1988). Co loss is generally controlled by biological uptake in oligotrophic regions (Moffett & Ho, 1996). Although the factors controlling Mn oxidation remain poorly understood (Lee & Fisher, 1993), Mn oxidation is the likely vector for Co scavenging given (1) the known ability for Co to be co-oxidized by Mn-oxidizing bacteria (Lee & Fisher, 1993; Moffett & Ho, 1996); (2) similar redox potentials and ionic radii of Co and Mn (Moffett & Ho, 1996; Swanner et al., 2014); (3) extensive covariation between Co and Mn contents of solid-phase marine sediments, manganese nodules, and ferromanganese crusts (Krishnaswami, 1976; Manheim, 1986), which accumulate Co scavenged from the water column; and (4) covariation of particulate Co and Mn phases in the mesopelagic (Saito et al., 2016). In addition to oxygen-related cycling of Mn-oxides in sediments, the absence of particulate Mn has been long noted in offshore oxygen minimum zones of the North and South Pacific (Johnson et al., 1996; Landing & Bruland, 1987; Ohnemus et al., 2017) and attributed to slow Mn oxide formation at low O_2 and in situ reduction. The specific rate of scavenging (A) is based on a minimum (ΛCo_{min}) and maximum scavenging rate (ΛCo) that is modulated by oxygen, bacterial activity (itself affected by nutrient and dissolved organic matter limitation), and light:

$$\Lambda = \Lambda_{MIN} + \Lambda Co \times Q \times kO_2 \times kBACT \times (1 - kPAR)$$
(7)

Q is the specific temperature function for Co oxidation by manganese-oxidizing bacteria with a Q_{10} of 2.75 (Lee & Fisher, 1993). The various other terms relate to the impact of oxygen, bacterial activity, and light on Co scavenging:

$$kO_{2} = \frac{(O_{2} - O_{25T})^{2}}{\left((O_{2} - O_{25T})^{2} + kO_{2}\Lambda^{2}\right)}$$
(8)

where O_2 is dissolved oxygen, O_{2ST} is the threshold concentration for scavenging, and $kO_2\Lambda$ is the half saturation constant for the influence of O_2 on Co scavenging.

$$kBACT = \frac{BACT^2}{(BACT^2 + kB\Lambda^2)}$$
(9)

BACT is the biomass of bacteria (μ M C; see Aumont et al., 2015) and kB Λ is the half saturation constant for the influence of bacterial activity on Co scavenging;



$$kPAR = \frac{PAR^2}{(PAR^2 + kPAR\Lambda^2)}$$
(10)

where PAR is photosynthetically active radiation and kPAR Λ is the half saturation constant for the influence of irradiance on Co scavenging. The overall loss of dCo (Scav_{Co}) is then governed by the scavenging rate (Λ) and the Co prime concentration (Co') such that Scav_{Co} = $\Lambda \times Co'$.

Since Cobinding ligands are very strong, with $\log K_{cond} > 16$ (Ellwood & van den Berg, 2001; Saito & Moffett, 2001; Saito et al., 2005), and are found at concentrations that are less than or equal to the dissolved Co concentration, we determine the Co prime (Co') concentration via Co' = dCo - CoL, where Co' is defined as the sum of inorganic cobalt complexes and Co²⁺. It should be noted that if there are any weaker organic Co complexes below the detection window of voltammetric studies, then they are part of the labile Co reservoir, which is a measured Co parameter that is often compared with Co'. The oxidation state of Co is not explicitly calculated within the model; however, empirical detection window studies have found that CoL complexes must be in the Co(III) state and solubility estimates require that Co' is Co(II) (Saito et al., 2005).

Co ligands have been observed to be produced by communities dominated by picocyanobacteria (Saito et al., 2005) and could be sourced from release of intracellular cobalamin/pseudocobalamin cofactors, or their precursors and photodegradation products, during the grazing or viral lysis of microbes in euphotic zone as part of the microbial loop. All sequenced marine cyanobacteria have the genes required for the de novo biosynthesis of pseudocobalamin, starting from inorganic Co species (Helliwell et al., 2016). We therefore link the production of Co ligands to the relative abundance and biomass of nanophytoplankton in our model, although future efforts could consider additional prokaryotic sources. Co ligands have an imposed minimum concentration (CoL_{MIN}) to stabilize dCo in the deep ocean. At present, this component of the model is a simple means to represent Co speciation and does not permit any excess Co binding ligands, although their presence remains debated (Ellwood et al., 2005; Saito & Moffett, 2001; Saito et al., 2005). The loss of dCo via scavenging is then Co' × Λ . At Co' concentrations greater than 100 pM, Co' is lost at an elevated rate (10 × Λ Co). Dissolution of scCo occurs where light is high or O₂ is low:

$$\lambda = \lambda_{MAX} \times MAX[kPAR, kO_2d] \tag{11}$$

where kPAR is as per equation (10) and

$$kO_2 d = \frac{(O_2 - O_{2DT})^2}{\left((O_2 - O_{2DT})^2 + kO_2 \Lambda^2 \right)}$$
(12)

where O_{2DT} is the threshold concentration for dissolution and λ_{MAX} is the specific rate of scCo dissolution. The specific rate of scCo dissolution is then multiplied by scCo to result in Dissol_{Co}.

2.3.3. Excretion and Remineralization

Zooplankton excretion of Co is modeled in the same manner as for Fe in PISCES-v2 and is accordingly enhanced when prey are rich in Co, relative to the imposed zooplankton Co quota (Table 1). The remineralization of organic Co, relative to organic P, can be upregualted or downregulated by the scalar ϕ . During model tuning, a 50% faster remineralization rate for Co, relative to P, was found to improve the Atlantic-Pacific contrast.

2.4. Model Experiments and Data Sets

The standard Co model (CTL) was spun up for 1,000 years, and a range of different experiments were then conducted for 125 years each. To determine the role of specific source processes, we ran experiments with no dust supply (NODUST) and no sediment supply (NOSED). In addition, the effect of oxygen thresholds on coastal sources and scavenging was investigated. We ran experiments where low oxygen did not enhance sediment Co fluxes (NOSEDOX; i.e., where $O_{2thres_1} = 0$) and where low oxygen did not switch off sedimentary Co fluxes (NOSEDOXA; i.e., where $O_{2thres_2} = 0$). We then ran a set of experiments where oxygen did not affect scavenging rates (SCAV1) and where bacterial activity did not affect scavenging rates (SCAV2).





Figure 1. Global cobalt data vertical distribution. Bars represent the standard deviation around the mean cobalt (pM) from each vertical bin (see methods). The blue line represents the model mean vertical profile extracted at the same location as the data while the red line represents the overall mean vertical profile of the model.

To assess our Co model, we compiled 8,235 Co data points from a variety of studies. They are compared graphically with the model results as raw data. For the statistical comparisons, the Co observations are gridded on to a $1^{\circ} \times 1^{\circ}$ horizontal grid with 33 vertical levels (bounded by 0, 10, 20, 30, 40, 50, 75, 100, 125, 150, 200, 250, 300, 400, 500, 600, 700, 800, 900, 1,000, 1,100, 1,200, 1,300, 1,400, 1,500, 1,750, 2,000, 2,500, 3,000, 3,500, 4,000, 4,500, 5,000, and 5,500 m) to compare directly with the model results on the same grid. We pay particular attention to examining the distributions of dCo along key GEOTRACES and CLIVAR transects in the Atlantic, Pacific, and Indian Oceans to evaluate model performance.

Data from the GEOTRACES GA03, CoFeMUG (GAc01), and GP16 sections are measured by cathodic stripping voltammetry following UV oxidation to destroy organic Co ligands. An analytical ligand, dimethyl glyoxime, binds to cobalt and is reduced with Co at a defined potential of -1.15 V, resulting in a reduction peak proportional to the Co concentration. Sample-specific instrument sensitivity is determined with four successive 25-pM standard additions, and a blank is subtracted. SAFe and GEOTRACES community standards are measured to ensure comparability with other methods.

Dissolved Co concentrations in samples from CLIVAR lines 18/19, P16, and 15 were measured using flowthrough solid phase extraction systems at FSU/NHMFL (Milne et al., 2010). For all cruises, samples were UV oxidized for 1.5 hr to destroy organic Co ligands and permit total extraction of Co by the chelating resin (Toyopearl AF-Chelate-650M for P16, Nobias Chelate PA-1 for 18/19 and 15). Sample aliquots (10–20 ml, 0.024 M HCl) were buffered to pH ~6 with ammonium acetate and flowed through a resin column at 2 ml/min. The captured Co was eluted from the resin using 0.5–1 ml of 1.0 M HNO₃ (UHP) and analyzed using high-resolution inductively coupled mass spectrometry (Thermo ELEMENT 2, NHMFL/FSU). Concentrations were quantified using standard additions, blank-corrected using aliquots of similarly extracted UHP water (18.2 M Ω cm), and verified through SAFe (S1 and D2) and GEOTRACES (GS and GD) community standards.

3. Results and Discussion

3.1. Modeled Co Distribution

We have compiled 8,235 dCo observations from the major ocean basins to serve as a basis for evaluating the skill of the base model. In general, the data displays the known "hybrid" character of dCo in the ocean (Figure 1). The surface ocean dCo levels are low on average but show high variability, while in the ocean interior dCo increases from the surface to intermediate depth but then declines from the intermediate to deep ocean (Figure 1). The mean model dCo profile extracted at the same location as the observations does a good job in reproducing the observed behavior, exhibiting a parallel decline in mean dCo toward the surface ocean and the ocean interior from intermediate water depth (~1,000 m), with a broadly similar pattern shown by the overall modeled mean dCo (Figure 1).

Spatially, the model is able to capture the major trends in dCo between different ocean regions and as a function of depth (Figure 2). As seen in observations, the Arctic Ocean is particularly dCo rich and declines with depth, whereas elsewhere dCo accumulates with depth, particularly so in the low oxygen regions of the tropical Atlantic, Pacific, and Indian Oceans. The major mismatch between the model and observations emerges in the Atlantic Ocean between 700–800 and 900–1,000 m, where the model underestimates the dCo levels within the low oxygen regions of the Mauritanian and Benguela upwelling areas. The CTL model has a correlation coefficient of >0.7 in the upper 200 m and >0.5 at depths greater than 900 m. The Atlantic mismatch between 700 and 800 m drives a correlation coefficient of 0.264 in this depth stratum (Table 3). Overall, the model has a correlation coefficient of 0.593 over the entire data set, which is comparable to the most skillful of global Fe models (Tagliabue et al., 2016).

Examining three GEOTRACES and three CLIVAR ocean sections reveals the strengths and weaknesses of the CTL model. On the GA-03 transect between Woods Hole and Cape Verde in the North Atlantic subtropical gyre (Noble et al., 2017), the model is able to represent a dCo maxima associated with the eastern and western margins, but these are too muted relative to the data (Figure 3a). This is likely driven by dCo removal rates that are too high, too little dCo supply from ocean margins, or an underestimation of regeneration of Co from sinking organic matter (Noble et al., 2017). A similar result is found for the CoFeMUG section across the south subtropical Atlantic (Noble et al., 2012) where the margin enhanced dCo is produced by the model but remains less widespread than the data (Figure 3b). In the Pacific Ocean, the model does an excellent job of reproducing the large dCo plume observed on the GP-16 transect (Hawco et al., 2016) and the low dCo upper ocean values (Figure 3c). The strong dCo maxima observed in the northern part of the Indian Ocean is well reproduced by the model along the CLIVAR I8 and I9 section (Figure 3d). The CLIVAR P16 section provides a unique window into the meridional distribution of dCo throughout the entire Pacific Ocean, and our model does a good job in reflecting the low dCo values in the surface ocean and gradual accumulation from south to north (Figure 3e). Finally, the zonal CLIVAR I05 section along the boundary between the Indian and Southern Oceans highlights the low dCo concentrations emanating from the dCo poor Southern Ocean (Figure 3f).

The skill of the model in the Atlantic Ocean is related to the underlying biogeochemical model rather than the Co submodel. Initial tests aimed at examining whether Atlantic dCo could be enhanced by lowering scavenging rates (Λ Co) or enhancing sedimentary dCo fluxes led to unrealistic accumulations of dCo in the Pacific and Indian basins. These initial tests led us to examine whether the biogeochemical model was overestimating oxygen levels in the Atlantic low oxygen regions, which then led to elevated scavenging rates (via the oxygen dependence of equation (7)). This was quantified by running an additional experiment where model oxygen is annually restored to World Ocean Atlas climatological values. In this run we observe a marked improvement in the modeled dCo along the GA-03 and CoFeMUG transects (Figure 4). Moreover, the model skill in the 700- to 800-m depth stratum is enhanced twofold (Table 2). This further emphasizes the importance of oxygen in shaping dCo cycling in the ocean interior. There is relatively little change in the modeled dCo sections along the GP-16 and CLIVAR I8 and I9 and P16 sections as the model already represents the low oxygen conditions well in these regions.

Our model also produces horizontal and vertical variations in the speciation of dCo. The greatest amounts of Co' (defined as the sum of inorganic Co complexes species and Co^{2+}) in both absolute and relative terms are found in the Arctic surface and interior waters and linked to the interior ocean oxygen minima of the tropical Atlantic, Pacific, and Indian oceans (Figure 5). These Co' distributions are consistent with some high latitude observations (Saito et al., 2010) and is in some part controlled by Co organic complexes being linked to the prevalence of nanophytoplankton in the model (based on cyanobacterial evidence; Saito et al., 2002), which are less prevalent at high latitudes. Scavenging removal is the other component driving the accumulation of Co'. In our model, Co' is removed by Mn-oxidizing bacteria, and it is only where this process is impeded that Co' can accumulate. In the oxygen minima, it is the low levels of oxygen that are restricting scavenging by Mn-oxidizing bacteria, while in the Arctic Ocean and to some extent the Southern Ocean, it is instead the cold temperatures that lessen the scavenging of Co' via lower bacterial metabolic rates. The ratio of Co' to dCo in the model is often greater than 0.5 in the high dCo plumes in the ocean interior, which is a slight overestimate relative to the available data (Bown et al., 2012; Hawco et al., 2016; Noble et al., 2012).

3.2. Role of External Sources

Our sensitivity experiments permit an evaluation as to the role of different Co source processes in different geographic regions. Unsurprisingly, dust supply of Co is most important in the regions of the ocean typified by significant dust deposition from the Sahara, Namibian, and Arabian deserts. Nevertheless, the largest effects found for the tropical Atlantic rarely exceed 5 pmol/L in the upper 50 m (Figure 6a), and eliminating the dust Co source (NODUST) does not greatly change the upper 50-m dCo from the CTL model (Figure 6b compared with Figure 2a). Dust Co is assumed to have a solubility of 8% (Shelley et al., 2012), so the muted influence of dust Co is mostly due to the low mineral fraction of Co in dust (17.3 μ g/g; Rudnick & Gao, 2014). These model results are consistent with observations and calculations showing small to nondetectable surface dust deposition effects in the Atlantic Ocean (Noble et al., 2012, 2017; Saito & Moffett, 2002; Shelley et al., 2012, 2016).





b) 100-200m



c) 400-500m



d) 700-800m



e) 900-1000m



Figure 2. Observed versus modeled dCo (annual mean, pM) for different depth slices.

f) 2500-3000m





Global Biogeochemical Cycles



Figure 3. Observed versus modeled dCo (annual mean, pM) for specific GEOTRACES and CLIVAR sections.



Global Biogeochemical Cycles



Figure 4. Observed versus modeled dCo (annual mean, pM) for two zonal Atlantic GEOTRACES sections when modeled oxygen is restored to World Ocean Atlas 2009 values.

In contrast to dust, sediments are the major external driver of dCo distributions at the surface and in the ocean interior, with absolute dCo concentrations modified by over 50 pM (Figures 6c and 6d). Indeed, in this experiment surface Co drops to very low levels <10 pM when sediment Co supply is eliminated (NOSED; Figure 6e). In our model, there are two major components to sediment supply; one is the base sediment Co flux, and the second is the enhancement of Co fluxes at low oxygen levels. When the enhanced Co fluxes at low oxygen are removed (NOSEDOX), we can highlight regions where Co supply from sediments is mostly driven by this particular process (Figures 6f and 6g) and the ensuing influence on surface dCo (Figure 6h). By then comparing the results of NOSED with NOSEDOX, we can calculate the percentage influence of low oxygen sediment fluxes on dCo. This calculation shows that in the tropical ocean, over half of the influence of sediment Co supply is governed by low oxygen enhancing fluxes, whereas in the Arctic ocean, the strong sensitivity to sediment Co fluxes is driven by the large shelf areas (Figures 6i and 6j).

Our model does not include hydrothermal input of dCo. While this could be included in the model in a similar manner as for Fe (Tagliabue & Resing, 2016), there is no evidence for large basin-scale dCo plumes alongside notable hydrothermal Fe signals (Hawco et al., 2016; Noble et al., 2012), despite observations of near-field localized sources (Noble et al., 2017, and references therein).

Statistical Evaluation of the model.							
	Standard model		World Ocean Atlas oxygen				
Depth interval (m)	R	RMSE	R	RMSE			
0–50	0.791	4.6	0.803	5.7			
100–200	0.735	5.0	0.800	5.0			
400-500	0.480	10.7	0.712	6.3			
700–800	0.264	23.6	0.578	7.0			
900-1,000	0.530	18.2	0.442	11.7			
2,500–3,000	0.607	13.2	0.478	11.3			
0–5,500	0.593	8.7	0.596	6.9			

Note. Correlation coefficient (*R*) and root-mean-square error (RMSE) for the standard model and a version of model where simulated oxygen is restored daily to World Ocean Atlas 2009 oxygen levels.

Table 2



a) Cobalt Prime (surface, pM)

b) Cobalt Prime / Dissolved Cobalt (surface, no units)





c) Cobalt Prime (250m, pM)





Figure 5. Free cobalt (pM) and the proportion of dCo that is free (no units) at the surface and at 250 m.

3.3. Role of Internal Cycling

We also conducted a set of sensitivity experiments examining the role of various removal processes affecting dCo in the ocean interior. In all these experiments, the Co loss due to a specific process was removed, allowing us to examine how a given process contributes to maintaining the modeled dCo levels.

To first consider oxygen, Figures 7a and 7b show the impact of eliminating the reduction in scavenging by low O_2 levels (equation (8)) at the surface and 250 m. In this experiment, scavenging rates are higher, and a clear effect in the tropical ocean emerges where low subsurface O_2 levels contribute upward of 50 pM to the dCo signal (Figure 7b). In regions where low O_2 zones in the ocean interior are coupled to the surface by upwelling, decreased Co scavenging at low O_2 increases surface ocean dCo by up to 10–20 pM (Figure 7a). In the oxygen-rich high-latitude oceans, there is little effect of low O_2 on lessening Co scavenging.

Eliminating the reduction in Co scavenging due to low levels of Mn-oxidizing bacterial activity (equation (9) has a different pattern to O_2 (Figures 7c and 7d). This parameterization for temperature-based controls on scavenging was based on laboratory experiments with Mn-oxidizing bacteria (Lee & Fisher, 1993) and field observations for limited dCo scavenging in the Ross Sea and under the sea-ice (Noble et al., 2013; Saito et al., 2010). The Arctic Ocean now emerges as the strongest signal, both at depth and at the surface (dCo declines by more than 80 pM when variations in bacterial rates are ignored). When linked to the strong role for sediment Co supply in this region (section 3.2), this indicates that the low rates of bacterial activity in these cold waters permit sedimentary Co to have a greater influence on dCo levels. The low latitude ocean is also impacted by the greater rates of Co scavenging when low levels of bacterial activity are eliminated. The impact of bacteria is broadly similar to O_2 , but much more widespread, both at the surface and at depth.





Figure 6. Absolute change in dCo (pM) at the surface (0–50 m) and at 250 m for no dust (a), no sediment supply (c and d), no sediment supply at low bottom water O_2 (f and g), and the percentage of the total change due to sediment supply caused by low bottom water O_2 sediment supply (i and j). (b, e, and h) The annually averaged surface dCo from the no dust, no sediment, and no sediment supply at low bottom water O_2 , respectively—please compare to Figure 2a.



100°E

60

20

7

3

-9

40

-80

a) ΔdCo , O₂ does not affect scavenging (surf, pM)

b) ΔdCo , O₂ does not affect scavenging (250m, pM)



c) ΔdCo , Bact does not affect scavenging (surf, pM)



0°

100°W



80°N

40°N

09

40°S

Figure 7. Absolute change in dCo (pM) at the surface and 250 m, when Co scavenging is not reduced at low oxygen levels (a and b) and not reduced by low rates of bacterial activity (c and d).

3.4. The Southern Equatorial Pacific Co Plume: A Case Study

The GEOTRACES GP16 cruise to the southern equatorial Pacific observed a notable offshore dCo plume in the subsurface ocean emanating from the Peru margin (Hawco et al., 2016). Our new Co model provides a way in which to assess how external input and internal cycling processes govern this high Co feature. As seen previously, the model is able to reproduce the intensity and magnitude of the observed plume (Figures 8a and 8b) better than the low oxygen associated Atlantic plumes (see above). We use our suite of sensitivity tests to quantify by how much the dCo plume declines when sedimentary Co supply and the decreased scavenging of Co driven by low O_2 and low rates of bacterial activity are removed. More than 70% of the dCo signal is eliminated by removing sedimentary Co supply very close to the margin, with the impact lessening further offshore (Figure 8c). The low O_2 enhancement of sedimentary Co fluxes supports ~25% of the dCo plume (Figure 8d). The impact of low rates of scavenging due to the low O_2 levels is muted very close to the margin but becomes much more important offshore (Figure 8e). This pattern is more marked for the role of low bacterial activity enhancing dCo levels, with little impact close to the margin but a greater impact offshore (Figure 8e). Thus, our model suggests that this dCo plume is initially controlled by high rates of scavenging, first due to low O_2 and then due to low rates of bacterial activity in the ocean interior.

4. A Synthesis of the Ocean Cobalt Cycle

We can use our model to bring together the first synthesis of the major external inputs and internal cycling of Co in the global ocean. Figure 9 shows the vertically integrated fluxes of Co due to dust and sediment supply (Figures 9a and 9b), biogeochemical processes of phytoplankton uptake and regeneration (Figures 9c and





c) ∆dCo/dCo, no sediment (GP-16, no units)



e) $\Delta dCo/dCo$, no O₂ scav (GP-16, no units)



b) Observed dCo (GP-16, pM)



d) $\Delta dCo/dCo$, no low O₂ sed (GP-16, no units)







Figure 8. How different processes shape the Co plume observed on GP-16. Top row shows (a) model and (b) data dCo (pM), while middle and bottom rows show the proportional change in dCo when there is (c) no sediment supply, (d) no enhanced sediment flux at low O_2 , (e) no reduction in scavenging at low O_2 , and (f) no reduction in scavenging at low rates of bacterial activity. Thick and thin contours highlight where a given process affects 50% and 25% of the magnitude of the dCo plume, respectively.





c) dCo consumption by biological uptake



e) dCo consumption by scavenging





d) dCo supply from pCo regeneration



f) dCo supply from scCo dissolution



Figure 9. Total fluxes of dCo linked to various external sources (a and b) and internal cycling (c-f). All fluxes are depth integrated and are presented in units of µmol Co/m² year.

9d), and the scavenging and dissolution of scavenged Co (Figures 9e and 9f). What becomes apparent is the strong influence of sediment fluxes at ocean boundaries that must then be transported widely by low interior scavenging rates. In our model, almost two thirds of the total global sedimentary boundary flux of Co is driven by our parameterization of enhanced supply when low bottom water oxygen is low. This points to a need for further studies on how bottom water oxygen levels modulate Co sediment supply. Co loss due to phytoplankton uptake and resupply due to regeneration are unsurprisingly associated with typical patterns of ocean biological productivity (Figures 9c and 9d). In a similar manner to the spatial coupling between Co consumption by biology and regeneration, Co scavenging and dissolution are spatially linked (Figures 9e and 9f). It is notable that low O_2 regions dissolve scavenged Co to dCo because of enhanced rates of scCo dissolution. This is not apparent in the higher O_2 regions of the high latitudes. In these regions, for example,





Figure 10. The magnitude of different processes in the modeled global cobalt budget (mol Co/year).

the Arctic, our model predicts that low temperatures decrease the activity of Mn-oxidizing bacteria yielding a lower scavenging removal of dCo.

Our model is able to provide the first estimates of the major global fluxes shaping the oceanic cycle of Co and the ocean residence time of Co (Figure 10 and Table 3). Dust, sediments, and rivers supply 6.5×10^7 , 6.8×10^8 , and 5.7×10^6 mol of Co annually. The sediment source compares favorably to an independent estimate ($\sim 6 \times 10^8$ mol of Co annually) based on simpler calculations from field data sets (Hawco et al., 2017). Primary production consumes 23.9×10^8 mol of Co, with much of this dCo sink balanced by recycling of 20.9×10^8 mol from zooplankton, while regeneration of particulate organic Co resupplies a further 8.6 \times 10⁸ mol each year. Globally, 3.1×10^8 and 2.5×10^8 mol of particulate Co (including organic and scavenged Co particles) sink across the 100- and 250-m depth horizons each year, respectively. Scavenging removes 6.8×10^8 mol from the dCo pool, and dissolution of scavenged Co returns 4.0×10^8 mol/year. When combined with the total Co inventory of the ocean in our model $(5 \times 10^{10} \text{ mol})$, the total Co inputs of 7.5×10^8 mol/year result in a global ocean Co residence time of 70 years. If this is split into upper 250 m and deeper than 250 m, then the resi-

dence times (ignoring physical exchanges) are approximately 7 years in the surface ocean and around 250 years deeper than 250 m (Table 3), similar to simpler early estimates (Bewers & Yeats, 1977; Saito & Moffett, 2002).

The internal cycling of Co at the global scale is driven by different processes between the surface ocean and the ocean interior. Unsurprisingly, biological uptake and Co turnover by zooplankton are the major Co sink and source terms in the upper 250 m $(23.9 \times 10^8 \text{ and } 20.7 \times 10^8 \text{ mol/year})$ where they dominate over the scavenging sink $(3 \times 10^8 \text{ mol/year})$. In addition to zooplankton recycling, resupply of dCo in the upper 250 m by particulate organic Co remineralization (8.5×10^8 mol/year) is around double that from the dissolution of scavenged Co (4×10^8 mol/year). Turning next to the ocean interior (>250 m), we find that dissolution of scavenged Co driven by low oxygen is the greatest dCo source globally $(3 \times 10^8 \text{ mol/year})$ and is more than four times greater than remineralization of particulate organic Co $(0.7 \times 10^8 \text{ mol/year, with zooplankton recycling})$ reduced to 0.2×10^8 mol/year). This switch in the dominant internal sources with depth is notable and may be unique to cobalt's biogeochemistry, with remineralization and recycling dominating in the upper water column and the dissolution of scavenged Co within the OMZs in the mesopelagic. Our emphasis on biological uptake in the upper 250 m agrees with a previous Co budget from the Atlantic Ocean (Dulaquais et al., 2014). However, in contrast to Dulaquais et al. (2014), we find dissolution of scavenged Co (putatively associated with Mn oxides) to be more important than organic Co remineralization in the ocean interior (deeper than 250 m) in our model. This difference likely reflects the fact that the work of Dulaguais et al. (2014) occurred in the relatively oxic Atlantic Ocean and dissolution of scCo will be much more important when the low oxygen

Table 3 Major Fluxes of the Modeled Cobalt Cycle		
External inputs	Internal cycling	Value
Dust		6.5×10^7 moles/year
Sediment		6.8×10^8 moles/year
River		$5.7 \times 10^{\circ}$ moles/year
Total		7.5×10^8 moles/year
	Primary production	24×10^8 moles/year
	Recycling	21×10^8 moles/year
	Regeneration	8.6×10^8 moles/year
	Scavenging	6.8×10^8 moles/year
	Dissolution	4.0×10^8 moles/year
	Sinking PCo (250 m)	2.5×10^8 moles/year
Global dCo inventory		5×10^{10} moles
Residence time (global)		250 years
Residence time (0-250 m)		7 years



Figure 11. Zonal mean dissolved cobalt (pM) from the Atlantic and Pacific Oceans. Zonal mean PO₄ (µM) is overlain as a contour.

zones of the Pacific and Indian Oceans are included (as in our global assessment). Our view is also consistent with observations of large dCo plumes within each of these major oxygen minimum zones (Hawco et al., 2016; Noble et al., 2012, 2017). Equally, it should be noted that particulate organic Co fluxes attenuate exponentially with depth, accounting for their greater importance in the upper 250 m and lesser role (in absolute terms) deeper than 250 m (Table 3). Finally, we highlight that these represent gross integrated fluxes from the model and a given Co atom may participate in more than one process during its lifetime in the ocean, for example, be remineralized from PCo, then scavenged to scCo and then dissolved back to dCo from scCo.

Our model has provided us with a conceptual view of how Co is transported from boundary sources into the ocean interior. The southern equatorial Pacific case study suggested that a strong source must be coupled with low scavenging rates to facilitate transfer throughout the ocean. The model experiments show that direct Co supply by dust is negligible apart from some very local regions in the tropical Atlantic Ocean. In the equatorial latitudes of the Pacific and also the Atlantic Ocean, low O₂ plays a key role in promoting Co transport by decreasing scavenging. This is seen by the imprint of high Co upon the meridional structure of the Atlantic and Pacific phosphate (PO₄) distributions at low latitudes (Figure 11). Additional decoupling between Co and PO₄ is observed at high latitudes in the Atlantic and Pacific. In the North Pacific, the model proposes an accumulation of dCo due to declining O_2 in the oldest waters at intermediate water depth (which has some support in CLIVAR data, Figure 3f, and other North Pacific data sets; M. Saito personal communication, 2017), while in the North Atlantic, high Co from the Arctic is transported equatorward (Figure 11). The Arctic is O₂ rich, compared to the low latitudes, and this region acts as a Co hotspot because high rates of Co input from the shallow shelves are coupled with low rates of removal due to cold temperatures depressing bacterial activity. In the Southern Ocean, shelves are narrower than in the Arctic, leading to lower Co input and little impact on dCo levels due to the scavenging loss in this highly oxic region. Ultimately, our model suggests that scavenging-dissolution processes and their modulation by oxygen levels and bacterial activity are the key determinants of the oceanic distribution of Co. Future studies characterizing the chemistry and biology of Co scavenging are warranted, in particular, the generation of in situ estimates of kinetic scavenging rates, to better constrain this process.

5. Toward Quantifying the Biological Role of Cobalt

In general, modeled phytoplankton Co quotas are lowest in the productive regions of the ocean and are highest in the oligotrophic gyres (Figures 12a and 12b). This reflects the fact that Co uptake in our model is





0-50m) b) Co/P diate

b) Co/P, diatoms. (µmol/mol, 0-50m)



c) Co/P, nanophyt. (μmol/mol, 0-50m, annual min)



e) Co/P, particles. (µmol/mol, 0-50m)





f) Co/P, particles. (μmol/mol, 1000-1500m)



Figure 12. Annual mean Co/P quotas in nanophytoplankton and diatoms at 0–50 m (a and b). Annual minimum Co/P quotas in nanophytoplankton and diatoms at 0–50 m using monthly model output (c and d). Annual mean Co/P ratios in all particles for 0–50 and 1,000–1,500 m (e and f). All are in units of μ mol Co/mol P.

independent of carbon and phosphorus (P) uptake, and thus, Co uptake can continue when growth rates (and C and P assimilation rates) are low. Due to the influence of Zn on Co uptake in diatoms, Co/P ratios are lowest for diatoms in the Zn-rich Southern Ocean. Over the seasonal cycle, nanophytoplankton and diatom Co/P quotas can reach the minimum values of ~60 and <10 μ mol/mol (Figures 12c and 12d) due to seasonal dCo depletion. Consistent with their overall low levels of phytoplankton biomass, absolute quantities of Co present in phytoplankton biomass are minimum in the oligotrophic gyres.

Modeled Co/P phytoplankton quotas reflect the observations compiled thus far from synchrotron X-ray fluorescence methods (Twining et al., 2011, 2015; Twining & Baines, 2013). These data sets find Co/P quotas in the temperate Pacific Ocean of <50 μ mol/mol for diatoms and >150 μ mol/mol for nondiatoms (King et al., 2012), while in the Equatorial Pacific Ocean, Co/P quotas are <100 μ mol/mol for diatoms and >150 μ mol/mol for nondiatoms (Twining et al., 2011), and the CTL model is able to reproduce these limited observations (Figures 12a–12d). The model finds that the subtropical North Atlantic Ocean displays elevated Co/P quotas for both diatoms and nanophytoplankton, and these are broadly reflected in the elevated cell quotas of 50– 170 μ mol/mol measured along GA-03 (Twining et al., 2015).

The Co/P ratio of the bulk particulate pool reflects the combination of the amalgamation of distinct assemblage quotas and any additional production of particulate Co via scavenging but does not include lithogenic Co. In



Relative impact of Zn availability on Co uptake

Figure 13. A map of the degree to which cobalt uptake is enhanced by Zn availability as per equation (6) (unitless).

general, the pattern (Figure 12e) represents that discussed previously for the phytoplankton. Co/P ratios are low (<100 μ mol/mol) in regions of high growth rate and in the Southern Ocean where the dominant diatom demand for Co is repressed by elevated Zn levels. In contrast, Co/P ratios are greatest (>150 μ mol/mol) in the tropical Atlantic and Indian Oceans. Notably this is without including any Co substitution within alkaline phosphatase of the dominant cyanobacteria populations in the model, which is suggested by observations of this metalloenzyme within regions of "accelerating" dCo:PO₄ stoichiometries (Saito et al., 2016). Finally, it is noteworthy that the Co/P ratios increase strongly with depth due to the production of additional particulate Co from the interior ocean scavenging of dCo by Mn-oxidizing bacteria (Figure 12f).

Ultimately, it is important to link the oceanic distributions and phytoplankton Co quotas to biological activity. At present, Co is known to have two major biological roles. First, vitamin B_{12} or cobalamin contains Co and is mainly required for the synthesis of the amino acid methionine and the nucleotide biosynthesis through the enzymes methionine synthase and ribonucleotide reductase, respectively (Bertrand et al., 2013; Rodionov et al., 2003). Second, it is known that Co can act as a substitute cofactor for Zn in carbonic anhydrase (Morel et al., 1994).

In our model, we accounted for the impact of Zn on Co requirements via equation (6). The precise degree of upregulation or downregulation of phytoplankton Co uptake is largely unknown due to variations in the diversity of Zn/Co cambialism; hence, its parameterization is relatively subjective at this stage. Nevertheless, the direction of change across the surface ocean is driven by Zn availability and should be relatively robust. Figure 13 displays the relative change in Co uptake due to Zn (equation (6)) and shows that maximum impact of Zn on Co requirements should be occurring in the oligotrophic gyres of the Pacific Ocean as Zn is depleted, followed by the southern subtropical Atlantic and the northern subtropical Atlantic gyres. Moreover, Zn-Co interactions may be further exacerbated in oligotrophic systems due to the connection between P scarcity and Zn/Co requirements that could explain high Co:P quotas in the surface Atlantic Ocean (Saito et al., 2016; Shaked et al., 2006). On the other hand, high levels of Zn in the Southern Ocean should lessen Co demands. Of course, this relies on the fact that we can broadly reconstruct Zn distributions from the close link between Zn and Si. In the future, it would be important to also develop a prognostic ocean Zn model that can be coupled to the current model.

At present, models such as PISCES do not account for vitamin regulation of phytoplankton physiology nor impact of Co or Zn scarcity affecting cellular enzymes. Instead, such global models tend to rely on identifying the most limiting resource that then governs carbon fixation rates. While some models are moving away from using the external nutrient concentration of resources to drive growth rates (Arteaga et al., 2014; Aumont et al., 2015), they still rely on a limited suite of resources and on "law of the minimum" parameterizations.



In the future, it is important for models to expand their scope beyond N, P, Si, and Fe to consider other important resources, such as Co, that are known to be depleted in seawater (Moore et al., 2013) and to revisit the resource limitation parameterizations to account for the potentially important colimitation between different resources. By way of an example, we found that diatom Co dropped markedly in the NOSED and NOSEDOX experiments, highlighting how remote sources from ocean boundaries supports Co nutrition and also implying that changes in boundary sources and their propagation into the ocean interior due to past or future climate change may affect Co limitation.

6. Conclusions

Overall, our model does a good job in reproducing the growing data set of dCo measurements arising from the GEOTRACES and CLIVAR efforts and allows for some of the first global-scale estimates of Co fluxes. We find an upper ocean residence time for Co of 7 years and a deep ocean residence time of 250 years, similar to previous estimates based on smaller data sets (Bewers & Yeats, 1977; Saito & Moffett, 2002). Our model highlights the sediments as the major external input of Co to the ocean and the importance of reduced scavenging removal in low oxygen regions such as the eastern tropical Pacific and cold regions such as the Arctic, in propagating Co throughout the ocean. The Arctic and Indian Oceans and low latitude upwelling systems are found to be the most Co-rich regions of the ocean, with the Southern Ocean and then the oligotrophic gyres as the most Co poor. Therefore, these Co-poor regions may be areas where Co has an impact on biological activity. Representing the impact of Co on microbial vital rates will, however, require a greater level of detail in the modeling of phytoplankton physiology in global models to account for resource substitution and colimitation. Such advances will shed important insights on metal quotas in marine phytoplankton given any future changes to external inputs and internal cycling of micronutrients.

Acknowledgments

We thank Gabriel Dulaguais and Maeve Lohan for providing published dissolved cobalt data sets from the Atlantic Ocean. This study is supported by funding from the European Research Council (project ID 724289), NERC (NE/ N001079/1), NSF OCE grants 1736599 and 1658030, and the Gordon and Betty Moore Foundation (3738). Collection of CLIVAR Co data used in this work was supported by four NSF OCE grants (0223378, 0649639, 0752832, and 0929919). A portion of this work was performed at the National High Magnetic Field Laboratory, which is supported by the National Science Foundation Cooperative Agreement no. DMR-1157490 and the State of Florida. We thank the two anonymous reviewers for their comments that improved the manuscript. Relevant model data are available from https://doi.org/10.5281/ zenodo.1196784.

References

- Arteaga, L., Pahlow, M., & Oschlies, A. (2014). Global patterns of phytoplankton nutrient and light colimitation inferred from an optimalitybased model. *Global Biogeochemical Cycles*, 28, 648–661. https://doi.org/10.1002/2013GB004668
- Aumont, O., Ethé, C., Tagliabue, A., Bopp, L., & Gehlen, M. (2015). PISCES-v2: An ocean biogeochemical model for carbon and ecosystem studies. Geoscientific Model Development, 8(8), 2465–2513. https://doi.org/10.5194/gmd-8-2465-2015
 - Bertrand, E. M., Saito, M. A., Rose, J. M., Riesselman, C. R., Lohan, M. C., Noble, A. E., et al. (2007). Vitamin B12 and iron colimitation of phytoplankton growth in the Ross Sea. *Limnology and Oceanography*, 52(3), 1079–1093. https://doi.org/10.4319/lo.2007.52.3.1079
 - Bertrand, E. M., Moran, D. M., McIlvin, M. R., Hoffman, J. M., Allen, A. E., & Saito, M. A. (2013). Methionine synthase interreplacement in diatom cultures and communities: Implications for the persistence of B12 use by eukaryotic phytoplankton. *Limnology and Oceanography*, 58(4), 1431–1450. https://doi.org/10.4319/lo.2013.58.4.1431
 - Bertrand, E. M., McCrow, J. P., Moustafa, A., Zheng, H., McQuaid, J. B., Delmont, T. O., et al. (2015). Phytoplankton-bacterial interactions mediate micronutrient colimitation at the coastal Antarctic Sea ice edge. *Proceedings of the National Academy of Sciences of the United of America*, 112(32), 9938–9943. https://doi.org/10.1073/pnas.1501615112

Bewers, J., & Yeats, P. (1977). Oceanic residence times of trace metals. *Nature, 268*(5621), 595–598. https://doi.org/10.1038/268595a0 Böning, P., Brumsack, H.-J., Böttcher, M. E., Schnetger, B., Kriete, C., Kallmeyer, J., & Borchers, S. L. (2004). Geochemistry of Peruvian near-

surface sediments. Geochimica et Cosmochimica Acta, 68(21), 4429–4451. https://doi.org/10.1016/j.gca.2004.04.027

Böning, P., Brumsack, H.-J., Schnetger, B., & Grunwald, M. (2009). Trace element signatures of Chilean upwelling sediments at ~36°S. Marine Geology, 259(1–4), 112–121. https://doi.org/10.1016/j.margeo.2009.01.004

Borchers, S. L., Schnetger, B., Böning, P., & Brumsack, H. J. (2005). Geochemical signatures of the Namibian diatom belt: Perennial upwelling and intermittent anoxia. Geochemistry, Geophysics, Geosystems, 6, Q06006. https://doi.org/10.1029/2004GC000886

- Bown, J., Boye, M., Baker, A., Duvieilbourg, E., Lacan, F., Le Moigne, F., et al. (2011). The biogeochemical cycle of dissolved cobalt in the Atlantic and the Southern Ocean south off the coast of South Africa. *Marine Chemistry*, 126(1–4), 193–206. https://doi.org/10.1016/j. marchem.2011.03.008
- Bown, J., Boye, M., & Nelson, D. M. (2012). New insights on the role of organic speciation in the biogeochemical cycle of dissolved cobalt in the southeastern Atlantic and the Southern Ocean. *Biogeosciences*, 9(7), 2719–2736. https://doi.org/10.5194/bg-9-2719-2012
- Bremner, J. M., & Willis, J. P. (1993). Mineralogy and geochemistry of the clay fraction of sediments from the Namibian continental margin and the adjacent hinterland. *Marine Geology*, 115(1-2), 85–116. https://doi.org/10.1016/0025-3227(93)90076-8

Browning, T. J., Achterberg, E. P., Rapp, I., Engel, A., Bertrand, E. M., Tagliabue, A., & Moore, C. M. (2017). Nutrient co-limitation at the boundary of an oceanic gyre. *Nature*, 551(7679), 242. https://doi.org/10.1038/nature24063

Bruland, K. W., Knauer, G. A., & Martin, J. H. (1978). Zinc in northeast Pacific water. Nature, 271(5647), 741–743. https://doi.org/10.1038/ 271741a0

Bruland, K. W., Middag, R., & Lohan, M. C. (2014). Controls of trace metals in seawater. In *Treatise on geochemistry* (2nd ed., pp. 19–51). https://doi.org/10.1016/b978-0-08-095975-7.00602-1

- Brumsack, H.-J. (1989). Geochemistry of recent TOC-rich sediments from the Gulf of California and the Black Sea. *Geologische Rundschau*, 78(3), 851–882. https://doi.org/10.1007/BF01829327
- Brumsack, H.-J. (2006). The trace metal content of recent organic carbon-rich sediments: Implications for cretaceous black shale formation. *Palaeogeography, Palaeoclimatology, Palaeoecology, 232*(2-4), 344–361. https://doi.org/10.1016/j.palaeo.2005.05.011

Cowen, J. P., & Bruland, K. W. (1985). Metal deposits associated with bacteria: Implications for Fe and Mn marine biogeochemistry. *Deep Sea Research Part A. Oceanographic Research Papers*, 32(3), 253–272. https://doi.org/10.1016/0198-0149(85)90078-0

- Droop, M. R. (1973). Some thoughts on nutrient limitation in algae. Journal of Phycology, 9(3), 264–272. https://doi.org/10.1111/j.1529-8817.1973.tb04092.x
- Droop, M. R. (1974). The nutrient status of algal cells in continuous culture. Journal of the Marine Biological Association of the United Kingdom, 54(04), 825. https://doi.org/10.1017/s002531540005760x
- Dulaquais, G., Boye, M., Middag, R., Owens, S., Puigcorbe, V., Buesseler, K., et al. (2014). Contrasting biogeochemical cycles of cobalt in the surface western Atlantic Ocean. *Global Biogeochemical Cycles*, 28, 1387–1412. https://doi.org/10.1002/2014GB004903
- Ellwood, M. J., & van den Berg, C. M. (2001). Determination of organic complexation of cobalt in seawater by cathodic stripping voltammetry. *Marine Chemistry*, 75(1-2), 33–47. https://doi.org/10.1016/S0304-4203(01)00024-X
- Ellwood, M. J., van den Berg, C. M. G., Boye, M., Veldhuis, M., de Jong, J. T. M., de Baar, H. J. W., et al. (2005). Organic complexation of cobalt across the Antarctic Polar Front in the Southern Ocean. *Marine and Freshwater Research*, 56(8), 1069–1075. https://doi.org/10.1071/ MF05097
- Gaillardet, J., Viers, J., & Dupré, B. (2003). Trace elements in river waters. Treatise on Geochemistry, 225–272. https://doi.org/10.1016/b0-08-043751-6/05165-3
- Hatta, M., Measures, C. I., Wu, J., Roshan, S., Fitzsimmons, J. N., Sedwick, P., & Morton, P. (2015). An overview of dissolved Fe and Mn distributions during the 2010–2011 U.S. GEOTRACES North Atlantic cruises: GEOTRACES GA03. Deep Sea Research Part II: Topical Studies in Oceanography, 116, 117–129. https://doi.org/10.1016/j.dsr2.2014.07.005

Hawco, N. J., Ohnemus, D. C., Resing, J. A., Twining, B. S., & Saito, M. A. (2016). A dissolved cobalt plume in the oxygen minimum zone of the eastern tropical South Pacific. *Biogeosciences*, 13(20), 5697–5717. https://doi.org/10.5194/bg-13-5697-2016

Hawco, N. J., Lam, P. J., Lee, J.-M., Ohnemus, D. C., Noble, A. E., Wyatt, N. J., et al. (2017). Cobalt scavenging in the mesopelagic ocean and its influence on global mass balance: Synthesizing water column and sedimentary fluxes. *Marine Chemistry*. https://doi.org/10.1016/j. marchem.2017.09.001

Heggie, D., & Lewis, T. (1984). Cobalt in pore waters of marine sediments. *Nature*, 311(5985), 453.

- Helliwell, K. E., Lawrence, A. D., Holzer, A., Kudahl, U. J., Sasso, S., Krautler, B., et al. (2016). Cyanobacteria and eukaryotic algae use different chemical variants of vitamin B12. *Current Biology*, 26(8), 999–1008. https://doi.org/10.1016/j.cub.2016.02.041
- Hu, D., Böning, P., Köhler, C. M., Hillier, S., Pressling, N., Wan, S., et al. (2012). Deep sea records of the continental weathering and erosion response to East Asian monsoon intensification since 14 ka in the south China Sea. *Chemical Geology*, 326-327, 1–18. https://doi.org/ 10.1016/j.chemgeo.2012.07.024
- Hu, D., Clift, P. D., Böning, P., Hannigan, R., Hillier, S., Blusztajn, J., et al. (2013). Holocene evolution in weathering and erosion patterns in the Pearl River delta. *Geochemistry, Geophysics, Geosystems, 14*, 2349–2368. https://doi.org/10.1002/ggge.20166
- Jickells, T. D., & Burton, J. D. (1988). Cobalt, copper, manganese and nickel in the Sargasso Sea. Marine Chemistry, 23(1–2), 131–144. https:// doi.org/10.1016/0304-4203(88)90027-8
- Johnson, K. S., Stout, P. M., Berelson, W. M., & Sakamoto-Arnold, C. M. (1988). Cobalt and copper distributions in the waters of Santa Monica Basin, California. *Nature*, 332(6164), 527–530. https://doi.org/10.1038/332527a0
- Johnson, K. S., Coale, K. H., Berelson, W. M., & Michael Gordon, R. (1996). On the formation of the manganese maximum in the oxygen minimum. *Geochimica et Cosmochimica Acta*, 60(8), 1291–1299. https://doi.org/10.1016/0016-7037(96)00005-1
- Kharkar, D., Turekian, K., & Bertine, K. (1968). Stream supply of dissolved silver, molybdenum, antimony, selenium, chromium, cobalt, rubidium and cesium to the oceans. *Geochimica et Cosmochimica Acta*, 32(3), 285–298. https://doi.org/10.1016/0016-7037(68)90016-1
- King, A. L., Sanudo-Wilhelmy, S. A., Boyd, P. W., Twining, B. S., Wilhelm, S. W., Breene, C., et al. (2012). A comparison of biogenic iron quotas during a diatom spring bloom using multiple approaches. *Biogeosciences*, 9(2), 667–687. https://doi.org/10.5194/bg-9-667-2012
- Krishnaswami, S. (1976). Authigenic transition elements in Pacific pelagic clays. *Geochimica et Cosmochimica Acta*, 40(4), 425–434. https://doi. org/10.1016/0016-7037(76)90007-7
- Landing, W. M., & Bruland, K. W. (1987). The contrasting biogeochemistry of iron and manganese in the Pacific Ocean. *Geochimica et Cosmochimica Acta*, 51(1), 29–43. https://doi.org/10.1016/0016-7037(87)90004-4
- Lee, B.-G., & Fisher, N. S. (1993). Microbially mediated cobalt oxidation in seawater revealed by radiotracer experiments. *Limnology and Oceanography*, 38(8), 1593–1602. https://doi.org/10.4319/lo.1993.38.8.1593
- Manheim, F. T. (1986). Marine cobalt resources. Science, 232(4750), 600-608. https://doi.org/10.1126/science.232.4750.600
- Martin, J. H., Gordon, R. M., Fitzwater, S., & Broenkow, W. W. (1989). Vertex—Phytoplankton iron studies in the Gulf of Alaska. Deep Sea Research, 36(5), 649. https://doi.org/10.1016/0198-0149(89)90144-1
- Milne, A., Landing, W., Bizimis, M., & Morton, P. (2010). Determination of Mn, Fe, Co, Ni, Cu, Zn, Cd and Pb in seawater using high resolution magnetic sector inductively coupled mass spectrometry (HR-ICP-MS). Analytica Chimica Acta, 665(2), 200–207. https://doi.org/10.1016/j. aca.2010.03.027
- Moffett, J. W., & Ho, J. (1996). Oxidation of cobalt and manganese in seawater via a common microbially catalyzed pathway. *Geochimica et Cosmochimica Acta*, 60(18), 3415–3424. https://doi.org/10.1016/0016-7037(96)00176-7
- Moore, C. M., Mills, M. M., Arrigo, K. R., Berman-Frank, I., Bopp, L., Boyd, P. W., et al. (2013). Processes and patterns of oceanic nutrient limitation. *Nature Geoscience*, 6(9), 701–710. https://doi.org/10.1038/ngeo1765
- Morel, F. M. M., Reinfelder, J. R., Roberts, S. B., Chamberlain, C. P., Lee, J. G., & Yee, D. (1994). Zinc and carbon co-limitation of marinephytoplankton. *Nature*, 369(6483), 740–742. https://doi.org/10.1038/369740a0
- Morse, J., & Luther, G. (1999). Chemical influences on trace metal-sulfide interactions in anoxic sediments. *Geochimica et Cosmochimica Acta*, 63(19-20), 3373–3378. https://doi.org/10.1016/S0016-7037(99)00258-6
- Noble, A. E., Saito, M. A., Maiti, K., & Benitez-Nelson, C. R. (2008). Cobalt, manganese, and iron near the Hawaiian islands: A potential concentrating mechanism for cobalt within a cyclonic eddy and implications for the hybrid-type trace metals. *Deep Sea Research Part II: Topical Studies in Oceanography*, 55(10–13), 1473–1490. https://doi.org/10.1016/j.dsr2.2008.02.010
- Noble, A. E., Saito, M. A., Moran, D. M., & Allen, A. (2013). Dissolved and particulate trace metal micronutrients under the McMurdo Sound seasonal sea ice: basal sea ice communities as a capacitor for iron. *Frontiers in Chemistry*, *1*, 25.
- Noble, A. E., Lamborg, C. H., Ohnemus, D. C., Lam, P. J., Goepfert, T. J., Measures, C. I., et al. (2012). Basin-scale inputs of cobalt, iron, and manganese from the Benguela-Angola front to the South Atlantic Ocean. *Limnology and Oceanography*, *57*(4), 989–1010. https://doi.org/ 10.4319/lo.2012.57.4.0989
- Noble, A. E., Ohnemus, D. C., Hawco, N. J., Lam, P. J., & Saito, M. A. (2017). Coastal sources, sinks and strong organic complexation of dissolved cobalt within the US North Atlantic GEOTRACES transect GA03. *Biogeosciences*, 14(11), 2715–2739. https://doi.org/10.5194/ bg-14-2715-2017
- Ohnemus, D. C., & Lam, P. J. (2015). Cycling of lithogenic marine particles in the US GEOTRACES North Atlantic transect. Deep Sea Research Part II: Topical Studies in Oceanography, 116, 283–302. https://doi.org/10.1016/j.dsr2.2014.11.019

- Ohnemus, D. C., Rauschenberg, S., Cutter, G. A., Fitzsimmons, J. N., Sherrell, R. M., & Twining, B. S. (2017). Elevated trace metal content of prokaryotic communities associated with marine oxygen deficient zones. *Limnology and Oceanography*, 62(1), 3–25. https://doi.org/10.1002/Ino.10363
 Price, N., & Marel, F. (2000). Conductive and exhibit view for view for view of a shark with view for view of a shark with view for view of a shark with view of a shark with view of a shark with view of view o
- Price, N., & Morel, F. (1990). Cadmium and cobalt substitution for zinc in a marine diatom. *Nature*, 344(6267), 658–660. https://doi.org/ 10.1038/344658a0
- Resing, J. A., Sedwick, P. N., German, C. R., Jenkins, W. J., Moffett, J. W., Sohst, B. M., & Tagliabue, A. (2015). Basin-scale transport of hydrothermal dissolved metals across the South Pacific Ocean. *Nature*, 523(7559), 200–203. https://doi.org/10.1038/nature14577
- Rodionov, D. A., Vitreschak, A. G., Mironov, A. A., & Gelfand, M. S. (2003). Comparative genomics of the vitamin B12 metabolism and regulation in prokaryotes. *The Journal of Biological Chemistry*, 278(42), 41148–41159. https://doi.org/10.1074/jbc.M305837200
- Rudnick, R. L., & Gao, S. (2014). Composition of the continental crust, 1–51. https://doi.org/10.1016/b978-0-08-095975-7.00301-6
 Saito, M. A., & Goepfert, T. J. (2008). Zinc–cobalt colimitation of *Phaeocystis antarctica*. *Limnology and Oceanography*, 53(1), 266–275. https://doi.org/10.4319/lo.2008.53.1.0266
- Saito, M. A., & Moffett, J. W. (2001). Complexation of cobalt by natural organic ligands in the Sargasso Sea as determined by a new highsensitivity electrochemical cobalt speciation method suitable for open ocean work. *Marine Chemistry*, 75(1-2), 49–68. https://doi.org/ 10.1016/S0304-4203(01)00025-1
- Saito, M. A., & Moffett, J. W. (2002). Temporal and spatial variability of cobalt in the Atlantic Ocean. Geochimica et Cosmochimica Acta, 66(11), 1943–1953. https://doi.org/10.1016/s0016-7037(02)00829-3
- Saito, M. A., Moffett, J. W., Chisholm, S. W., & Waterbury, J. B. (2002). Cobalt limitation and uptake in Prochlorococcus. Limnology and Oceanography, 47(6), 1629–1636. https://doi.org/10.4319/lo.2002.47.6.1629
- Saito, M. A., Sigman, D. M., & Morel, F. M. M. (2003). The bioinorganic chemistry of the ancient ocean: The co-evolution of cyanobacterial metal requirements and biogeochemical cycles at the Archean–Proterozoic boundary? *Inorganica Chimica Acta*, 356, 308–318. https://doi. org/10.1016/S0020-1693(03)00442-0
- Saito, M. A., Moffett, J. W., & DiTullio, G. R. (2004). Cobalt and nickel in the Peru upwelling region: A major flux of labile cobalt utilized as a micronutrient. *Global Biogeochemical Cycles*, 18, GB4030. https://doi.org/10.1029/2003GB002216
- Saito, M. A., Rocap, G., & Moffett, J. W. (2005). Production of cobalt binding ligands in a Synechococcus feature at the Costa Rica upwelling dome. Limnology and Oceanography, 50(1), 279–290. https://doi.org/10.4319/lo.2005.50.1.0279
- Saito, M. A., Goepfert, T. J., & Ritt, J. T. (2008). Some thoughts on the concept of colimitation: Three definitions and the importance of bioavailability. *Limnology and Oceanography*, 53(1), 276–290. https://doi.org/10.4319/lo.2008.53.1.0276
- Saito, M. A., Goepfert, T. J., Noble, A. E., Bertrand, E. M., Sedwick, P. N., & DiTullio, G. R. (2010). A seasonal study of dissolved cobalt in the Ross Sea, Antarctica: Micronutrient behavior, absence of scavenging, and relationships with Zn, Cd, and P. *Biogeosciences*, *7*(12), 4059–4082. https://doi.org/10.5194/bg-7-4059-2010
- Saito, M. A., Noble, A. E., Tagliabue, A., Goepfert, T. J., Lamborg, C. H., & Jenkins, W. J. (2013). Slow-spreading submarine ridges in the South Atlantic as a significant oceanic iron source. *Nature Geoscience*, 6(9), 775–779. https://doi.org/10.1038/Ngeo1893
- Saito, M. A., Noble, A., Hawco, N., Twining, B. S., Ohnemus, D. C., John, S. G., et al. (2016). The acceleration of dissolved cobalt's ecological stoichiometry due to biological uptake, remineralization, and scavenging in the Atlantic Ocean. *Biogeosciences Discussions*, 1–43. https:// doi.org/10.5194/bq-2016-511
- Sañudo-Wilhelmy, S. A., Gobler, C. J., Okbamichael, M., & Taylor, G. T. (2006). Regulation of phytoplankton dynamics by vitamin B12. Geophysical Research Letters, 33, L04604. https://doi.org/10.1029/2005GL025046
- Shaked, Y., Xu, Y., Leblanc, K., & Morel, F. M. M. (2006). Zinc availability and alkaline phosphatase activity in *Emiliania huxleyi*: Implications for Zn-P co-limitation in the ocean. *Limnology and Oceanography*, *51*(1), 299–309. https://doi.org/10.4319/lo.2006.51.1.0299
- Shelley, R. U., Sedwick, P. N., Bibby, T. S., Cabedo-Sanz, P., Church, T. M., Johnson, R. J., et al. (2012). Controls on dissolved cobalt in surface waters of the Sargasso Sea: Comparisons with iron and aluminum. *Global Biogeochemical Cycles*, 26, Gb2020. https://doi.org/10.1029/ 2011GB004155
- Shelley, R. U., Wyatt, N. J., Tarran, G. A., Rees, A. P., Worsfold, P. J., & Lohan, M. C. (2016). A tale of two gyres: Contrasting distributions of dissolved cobalt and iron in the Atlantic Ocean during an Atlantic Meridional Transect (AMT-19). Progress in Oceanography, 158, 52–64. https://doi.org/10.1016/j.pocean.2016.10.013
- Sunda, W. G., & Huntsman, S. A. (1988). Effect of sunlight on redox cycles of manganese in the southwestern Sargasso Sea. Deep Sea Research, 35(8), 1297–1317. https://doi.org/10.1016/0198-0149(88)90084-2
- Sunda, W. G., & Huntsman, S. A. (1995). Cobalt and zinc interreplacement in marine phytoplankton: Biological and geochemical implications. Limnology and Oceanography, 40(8), 1404–1417. https://doi.org/10.4319/lo.1995.40.8.1404
- Sundby, B., Anderson, L. G., Hall, P. O., Iverfeldt, Å., van der Loeff, M. M. R., & Westerlund, S. F. (1986). The effect of oxygen on release and uptake of cobalt, manganese, iron and phosphate at the sediment-water interface. *Geochimica et Cosmochimica Acta*, 50(6), 1281–1288. https://doi.org/10.1016/0016-7037(86)90411-4

Swanner, E. D., Planavsky, N. J., Lalonde, S. V., Robbins, L. J., Bekker, A., Rouxel, O. J., et al. (2014). Cobalt and marine redox evolution. *Earth and Planetary Science Letters*, 390, 253–263. https://doi.org/10.1016/j.epsl.2014.01.001

- Tagliabue, A., & Resing, J. A. (2016). Impact of hydrothermalism on the ocean iron cycle. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences,* 374(2081), 20150291. https://doi.org/10.1098/rsta.2015.0291
- Tagliabue, A., Aumont, O., DeAth, R., Dunne, J. P., Dutkiewicz, S., Galbraith, E., et al. (2016). How well do global ocean biogeochemistry models simulate dissolved iron distributions? *Global Biogeochemical Cycles*, *30*, 149–174. https://doi.org/10.1002/2015GB005289
- Twining, B. S., & Baines, S. B. (2013). The trace metal composition of marine phytoplankton. *Annual Review of Marine Science*, 5(1), 191–215. https://doi.org/10.1146/annurev-marine-121211-172322
- Twining, B. S., Baines, S. B., Bozard, J. B., Vogt, S., Walker, E. A., & Nelson, D. M. (2011). Metal quotas of plankton in the equatorial Pacific Ocean. *Deep-Sea Res Pt II*, 58(3–4), 325–341. https://doi.org/10.1016/j.dsr2.2010.08.018
- Twining, B. S., Rauschenberg, S., Morton, P. L., & Vogt, S. (2015). Metal contents of phytoplankton and labile particulate material in the North Atlantic Ocean. *Progress in Oceanography*, 137, 261–283. https://doi.org/10.1016/j.pocean.2015.07.001
- Wojciechowski, C. L., Cardia, J. P., & Kantrowitz, E. R. (2002). Alkaline phosphatase from the hyperthermophilic bacterium *T. maritima* requires cobalt for activity. *Protein Science*, *11*(4), 903–911. https://doi.org/10.1110/ps.4260102
- Xu, Y., Tang, D., Shaked, Y., & Morel, F. M. (2007). Zinc, cadmium, and cobalt interreplacement and relative use efficiencies in the coccolithophore *Emiliania huxleyi*. *Limnology and Oceanography*, 52(5), 2294–2305. https://doi.org/10.4319/lo.2007.52.5.2294
- Zhang, H., Van Den Berg, C. M., & Wollast, R. (1990). The determination of interactions of cobalt (II) with organic compounds in seawater using cathodic stripping voltammetry. *Marine Chemistry*, 28(4), 285–300. https://doi.org/10.1016/0304-4203(90)90049-I